Author’s response to reviews

Title: Evaluation of the impact of a brief educational message on clinicians’ awareness of risks of ionising-radiation exposure in imaging investigations: a pilot pre-post intervention study

Authors:

Ben Young (ben.young@nottingham.ac.uk)
Jo Cranwell (j.cranwell@bath.ac.uk)
Andrew Fogarty (andrew.fogarty@nottingham.ac.uk)
Rob Skelly (rob.skelly@nhs.net)
Nigel Sturrock (nigel.sturrock1@nhs.net)
Mark Norwood (mark.norwood@nhs.net)
Dominick Shaw (dominic.shaw@nottingham.ac.uk)
Sarah Lewis (sarah.lewis@nottingham.ac.uk)
Tessa Langley (tessa.langley@nottingham.ac.uk)
Peter Thurley (peter.thurley2@nhs.net)

Version: 2 Date: 31 Oct 2019

Author’s response to reviews:

We have made the requested revisions.